Author's response to reviews

Title: The TERT rs2736100 Polymorphism and Cancer Risk: A Meta-analysis Based on 25 Case-Control Studies

Authors:

Peng Zou (isdwhzp@126.com)
Aihua Gu (aihuagu@njmu.edu.cn)
Guixiang Ji (gxji@njmu.edu.cn)
Lin Zhao (postletterplant@163.com)
Peng Zhao (zhaopeng@njmu.edu.cn)
Ailin Lu (luailin.eric@163.com)

Version: 3 Date: 3 December 2011

Author's response to reviews: see over
Dear Editors and Reviewers,

We highly appreciate the detailed valuable suggestion on our manuscript of ‘MS 102907895692742 (The TERT rs2736100 Polymorphism and Cancer Risk: A Meta-analysis Based on 25 Case-Control Studies)’. The suggestion is quite helpful for us. We have modified the manuscript accordingly, and the detailed corrections are listed below.

We hope the Reviewers and the Editors will be satisfied with our responses to the ‘comments’ and the revisions for our manuscript. We’re happy that one reviewer (Dr. Hemant Bid) has accepted our revised manuscript.

Thanks and Best Regards!

Yours sincerely,

Peng Zhao

Responses to the quests:

Reviewer 1: Dr. Kshitij Srivastava

Minor Essential Revisions
1. The sentence "In our meta-analysis, most of the cancer types were lung cancer and glioma" in background section of manuscript should be removed and put in results section.

Many thanks for the wonderful advice. We have revised it as suggested. The sentence "In our meta-analysis, most of the cancer types were lung cancer and glioma" have been removed and put in the first paragraph of results section and highlighted with yellow background color.